I2P Guide: C. Andy Tsao Notes and Homework

Week 3. Fall 2002.

Outline

In a simple sample space, each outcome (in the sample space) has equal probability to occur. Therefore, the assignment of a probability frequently transforms into a counting problem: how many possible cases are there. This gives good reasons to learn about counting methods. 1. Counting Methods

(a) Multiplication Rule. General version of Multiplication Rule (cf. Note at the bottom of P23) (b) Permutations, Combinations and connection between them. (c) Sampling with or without replacement (d) Ordering (e) Binomial, Multinomial coefficient; Binomial, Multinomial Theorem. Binomial as a special case of Multinomial.

2. Famous Problems: Birthday problem, Problem of Division (Problem of Points), Tennis Problem.

Homework (due 021004)

1. §1.7: 1, 4, 10. §1.8: 3, 7, 20 §1.9: 4, 9, 11.

Learning Guide and Announcement

1. Skim through Chapter 1–2, Intro and Summary in case you haven’t done so. 2. Review §1.1–§1.9. Preview §1.10–§1.11.

3. Prepare for Quiz 1 .

4. The TA’s and their office hours will be announced shortly.

Quote of the Week

Who asks is an idiot for 5 minutes, who never asks stays an idiot for life. ♠♥♦♣

Page 1 of 1 Typeset by AMS-LaTeX